Author's response to reviews

Title: The efficacy of computer reminders on external quality assessment for point-of-care testing in Danish general practice: Rationale and methodology for two randomised trials

Authors:

Frans B Waldorff (fransw@sund.ku.dk)
Volkert Siersma (siersma@sund.ku.dk)
Ruth Ertmann (ruth@sund.ku.dk)
Marius Brostrøm Kousgaard (marbro@sund.ku.dk)
Anette Sonne Nielsen (anette.sonne.nielsen@regionh.dk)
Peter Felding (pf@kpll.dk)
Niels Mosbæk (nmo@kpll.dk)
Else Hjortsoe (else.hjortsoe@regionh.dk)
Susanne Reventlow (susrew@sund.ku.dk)

Version: 8 Date: 15 July 2011

Author's response to reviews:

Dear editorial team, please see below. Best Frans

Formatting changes to be made:
-----------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Done

Figures: We notice that you have uploaded Figures 1 and 2 both as Figure 1. Please upload the two figures as separate files, in numerical order.
Done

Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: please include the city with Affiliations 3 and 4. Done

Blank Pages: please remove any blank pages from the manuscript. Done
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures. Uncertain.

Figures: please remove the footnotes at the bottom of Figure 1 and place them in the figure legend within the main text instead. Done